Author’s response to reviews

Title: Determining the Impact of Smoking Point of sale Legislation Among Youth (DISPLAY) Study: a protocol for an evaluation of public health policy

Authors:

Sally Haw (s.j.haw@stir.ac.uk)
Amanda Amos (amanda.amos@ed.ac.uk)
Douglas Eadie (douglas.eadie@stir.ac.uk)
John Frank (john.frank@ed.ac.uk)
Laura MacDonald (l.e.macdonald@stir.ac.uk)
Anne Marie MacKintosh (a.m.mackintosh@stir.ac.uk)
Andy MacGregor (andy.macgregor@scotcen.org.uk)
Martine Miller (mam30@st-andrews.ac.uk)
Jamie Pearce (jamie.pearce@ed.ac.uk)
Clare Sharp (clare.sharp@scotcen.org.uk)
Martine Stead (martine.stead@stir.ac.uk)
Catherine Tisch (ctisch@staffmail.ed.ac.uk)
Winfried van der Sluijs (wpmvds@st-andrews.ac.uk)

Version: 2 Date: 25 October 2013

Author’s response to reviews: see over
5th September, 2013

Dear Editor

Determining the Impact of Smoking Point of sale Legislation Among Youth (DISPLAY)
Study: a protocol for an evaluation of public health policy

Further to your recent email, please find attached a revised version of the above paper. The changes are as follows:

Page 1: Word count now 4,900
Page 15: New section, entitled Sample Size and Power Calculation for School Surveys (this was omitted in the first submission due to length but after reconsideration of the paper I decided to include it.)
Page 21: Insert of section on Ethical Approval
Page 23: Insert of references 18-20
Page 27: Insert of table 3
Page 28: Insert of table 4

Yours sincerely

Sally Haw
Professor of Public & Population Health
Centre for Public & Population Health